\documentclass[12pt]{article}
\usepackage{amsmath}
\setlength\parindent{0pt}

%PAGE LAYOUT
\pagestyle{empty}
\setlength{\topmargin}{-1in}
\setlength{\headheight}{1.5cm}
\setlength{\headsep}{0.3cm}
\setlength{\textheight}{9in}
\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{0cm}
\setlength{\evensidemargin}{0cm}

%\usepackage{graphicx}

\begin{document}

\begin{tabular}{@{}l p{4.5cm} r@{}}
	Name: \hrulefill & & Math 1441 Quiz 9 \\[0.2cm]
	\hspace{8cm} & & 
\end{tabular}

{\bf 1.} (10 points) Use \textbf{areas} to evaluate the integral.
\begin{flalign}
&(a)\ \int_{-1}^1 7 \,dx&
&(b)\ \int_{-1}^1 |x| \,dx&
&(c)\ \int_{-1}^1 -\sqrt{1-x^2} \,dx\notag
\end{flalign}\\[8cm]

{\bf 2.} (10 points) Use the \textbf{Fundamental Theorem of Calculus} to evaluate the integral.
\begin{flalign}
	&(a)\ \int_0^4 {\sqrt{x}} \,dx&
	&(b)\ \int_1^2 \frac{1}{x^2} \,dx&
	&(c)\ \int_{0}^1 x(x-1) \,dx\notag
\end{flalign}

\end{document}